Author's response to reviews

Title: Prevalence and Perceived Health Effect of Alcohol use among Male Undergraduate Students in Owerri, South-East Nigeria.

Authors:

Chikere I.C Ebirim (chayke2000@yahoo.com)
Mayowa O. Morakinyo (wahlemirax@googlemail.com)

Version: 4 Date: 19 January 2011

Author's response to reviews: see over
Title: Prevalence and Perceived Health Effect of Alcohol use among Male Undergraduate Students in Owerri, South-East Nigeria.

Authors:

Chikere I.C Ebirim (chayke2000@yahoo.com)

Mayowa O. Morakinyo (wahlemirax@gmail.com)

Version: 4 Date: 13 January 2011

Comments: see over
Cover Letter

Title: Prevalence and Perceived Health Effect of Alcohol use among Male Undergraduate Students in Owerri, South-East Nigeria: a descriptive cross-sectional study.

Associate Editor Requirements:

Comment 1) remark: the correction has been effected.
Comment 2) remark: manuscript has been revised.
Comment 3) remark: correction effected.

I appreciate your concerted effort in the review of this manuscript, i want to say am delighted to hear that the manuscript has been accepted.

Thank you.